Author's response to reviews

Title: Haiti and the Health Marketplace: The Results are Perishable

Authors:

Jo Durham (durhamjo@yahoo.com)
Markus J Michael (marcos.j.michael@gmail.com)
Peter S Hill (p.hill@sph.uq.edu.au)
Enrico Pavignani (enricopavignani@hotmail.com)

Version: 2 Date: 30 May 2013

Author's response to reviews: see over
30th May 2013

Dear Editor

Please find attached the manuscript: Haiti and the Health Marketplace: The Results are Perishable. We did not realise the initial version as submitted as I not read the pdf. We have now made the required corrections as below:

1. We have added a statement about competing interests:

   The authors declare that they have no competing interests

2. Authors' Contributions

   We have added a statement about the Authors' Contributions section after Competing interests:

   PSH, MM, EP participated in the design of the study. PSH and MM carried out the qualitative interviews and analysis. JD contributed to the analysis and drafted the manuscript. All authors contributed to and approved the final manuscript.

3. Consent statement

   Please state in the Methods section whether written informed consent for participation in the study was obtained from participants or, where participants are children, a parent or guardian. We have added:

   All interviews were undertaken with oral consent, including for recording, following witnessed reading of information sheet.

4. We have used the template provided

   We believe the fits the interest of you readership especially given the growing literature on health in fragile states and severely disrupted environments. All authors have contributed to and read the paper. It has not been submitted elsewhere. The study was approved by Ethics approval for the research was obtained from the University of Queensland Research Ethics Committee. The Committee reviewed the research protocols, interview guides, informed consent process and the consent and information forms to be used. The Ministry of Public Health and Population in Haiti was provided with the research protocol and informed of the ethics approval.

   Thank you for your consideration and we look forward to hearing from you.
Yours sincerely

Jo Durham